

Geophysical Research Letters[•]

RESEARCH LETTER

10.1029/2022GL098552

Key Points:

- Warming from climate change accelerates runoff and decreases streamflow during low flow periods in many snow dependent regions
- Extended summer low flows reduce and fragment the capacity of freshwater habitats to support historical salmon abundances
- Climate adaptation actions for salmon should consider hydrologic changes and floodplain connectivity besides increasing water temperatures

Supporting Information:

Supporting Information may be found in the online version of this article.

Correspondence to:

D. Tonina, dtonina@uidaho.edu

Citation:

Tonina, D., McKean, J. A., Isaak, D., Benjankar, R. M., Tang, C., & Chen, Q. (2022). Climate change shrinks and fragments salmon habitats in a snowdependent region. *Geophysical Research Letters*, 49, e2022GL098552. https://doi. org/10.1029/2022GL098552

Received 2 MAR 2022 Accepted 17 MAY 2022

Author Contributions:

Conceptualization: Daniele Tonina, James A. McKean, Daniel Isaak, Qiuwen Chen

Data curation: James A. McKean Formal analysis: Daniele Tonina, James A. McKean, Daniel Isaak, Rohan M. Benjankar, Chunling Tang, Qiuwen Chen Funding acquisition: James A. McKean, Daniel Isaak Project Administration: Daniele Tonina, James A. McKean Validation: Rohan M. Benjankar Visualization: Daniele Tonina, James A. McKean, Chunling Tang Writing – original draft: James A. McKean, Daniel Isaak Writing – review & editing: Daniele Tonina, James A. McKean, Daniel Isaak, Rohan M. Benjankar Chunling Tang

Rohan M. Benjankar, Chunling Tang, Qiuwen Chen

© 2022. American Geophysical Union. All Rights Reserved.

Climate Change Shrinks and Fragments Salmon Habitats in a Snow-Dependent Region

Daniele Tonina¹ ^(b), James A. McKean² ^(b), Daniel Isaak², Rohan M. Benjankar³ ^(b), Chunling Tang⁴ ^(b), and Qiuwen Chen⁵ ^(b)

¹Center for Ecohydraulics Research, University of Idaho, Boise, ID, USA, ²Rocky Mountain Research Station, Aquatic Laboratory, Boise, ID, USA, ³Southern Illinois University Edwardsville, Edwardsville, IL, USA, ⁴US Environmental Protection Agency, Washington, DC, WA, USA, ⁵Nanjing Hydraulics Research Institute, Nanjing, China

Abstract Climate change threatens biodiversity through global alteration of habitats, but efficient conservation responses are often hindered by imprecise downscaling of impacts. Besides thermal effects, warming also drives important ancillary environmental changes, such as when river hydrology evolves in response to climate forcing. Earlier snowmelt runoff and summer flow declines are broadly manifested in snow-dependent regions and relevant to socioeconomically important cold-water fishes. Here, we mechanistically quantify how climate-induced summer flow declines during historical and future periods cause complex local changes in Chinook salmon (*Oncorhynchus tshawytscha*) habitats for juveniles and spawning adults. Changes consisted of large reductions in useable habitat area and connectivity between the main channel and adjacent off-channel habitats. These reductions decrease the capacity of freshwater habitats to support historical salmon abundances and could pose risks to population persistence in some areas.

Plain Language Summary The large majority of climate change research for aquatic organisms has focused on stream water temperature and discounted the importance of flow changes associated with evolving precipitation regimes. Using a high-resolution stream topographic map that supported a series of linked flow and fish habitat quality models, our research describes how observed historical flow reductions and projected future reductions affect local habitat quality, distributions, and connectivity for a salmon population. We demonstrated that as climate change continues to change flow regimes in snow-dependent regions, it will cause fragmentation and reduce the overall volume of available habitats in headwater streams where salmon and other cold-water fish species spawn and grow during early life stages.

1. Introduction

A changing climate pervasively affects life on Earth (Parmesan & Yohe, 2003) and is projected to drive to extinction many local populations and some taxa this century (Urban, 2015) and threaten the habitat of many cold-water fish such as salmonids (Isaak et al., 2018; Young et al., 2018). It leads to changes in temperature and hydrology with biological consequences that are complicated by complex interactions with local habitats and physical conditions (Wenger et al., 2011). Despite progress in recent decades, we still lack the ability to represent local climatic conditions at the resolutions needed for greatest relevance in ecological risk assessments and strategic application of conservation resources (Potter et al., 2013). Challenges are magnified in stream and river environments that are disproportionately biodiverse (Dudgeon et al., 2006), but difficult to study given their dynamic physical properties and problems associated with making measurements in flowing water. Emergence of new remote sensing (McKean et al., 2009) and hydrodynamic modeling technologies (Nelson et al., 2016), however, have enabled bathymetric surveys and analyses of lotic environments to progress rapidly in recent years and applications are proliferating (Carbonneau et al., 2012).

In the Columbia River Basin of the northwestern U.S., rivers provide habitats for world-renowned salmon populations and fisheries but the abundance of most species has declined precipitously due to habitat degradation, fragmentation, and overharvest (Nehlsen et al., 1991). This has prompted extensive restoration efforts in hundreds of rivers and streams and cost billions of US\$ in previous decades (Lewis et al., 2015; Rieman et al., 2015). Hydrologic regimes within the region are snow-dependent and climate-sensitive, similar to most areas inhabited by salmon throughout North America and Eurasia because of salmon species' requirement for cold riverine environments. Increasing air temperatures throughout the northwest in recent decades are linked to reductions in winter snow accumulations, earlier spring snowmelt runoff, and declining summer flows (Isaak et al., 2018; Kormos et al., 2016). The summer low flow period is an important bottleneck for most salmon populations that are territorial during juvenile and adult life stages (Quinn, 2005) because reductions in water volume enhance resource competition and translate to decreased habitat capacity and population abundance (Ohlberger et al., 2018). If decreases in habitat extent are accompanied by reductions in flow velocities or even stagnant backwater areas with no flow, remaining habitats are also likely to become less productive for juvenile salmon, which feed on passively drifting insects that are delivered at rates commensurate with flow velocities (Naman et al., 2016).

Here, we investigate the effects of climate-induced summer flow reductions on salmon spawning and rearing habitats in both in-channel and off-channel habitats (OCH). To address our goal, we surveyed a typical Chinook salmon spawning stream and adjacent floodplain, at Bear Valley Creek (Idaho, US, Figure 1a), using an aircraft-mounted, water penetrating green LiDAR system (McKean et al., 2009). A detailed 1 m² resolution bathymetry (Figure 1b) was developed from the survey data and used to define channel boundary conditions for hydrody-namic models that quantified local flow velocity (Figure 1c) and habitat characteristics (Figures 1d and 1e) associated with historical and future summer discharge levels along a 7.4 km reach (Study site hydrology, Figure S1 in Supporting Information S1).

2. Study Site

Bear Valley Creek is a tributary to the Middle Fork of the Salmon River (Idaho, USA; Figure 1a). It is typical of many headwater streams used by salmon for adult spawning and juvenile rearing throughout the Pacific Northwest, USA. It is a small (bankfull width of 15 m), meandering (1.5 sinuosity index), low-gradient (0.0035 m/m), gravel bed (median grain size, d_{50} , of 0.052 m) stream that flows in a broad, gentle valley formed in glacial outwash deposits (McKean et al., 2008; McKean and Tonina, 2013; Schmidt & Mackin, 1970). The stream is largely unconfined in the study reach with small discontinuous terrace remnants breaking up the valley surface that is also marked with numerous abandoned and semi-abandoned channel fragments, sloughs, and shallow closed depressions that constitute the off-channel habitat when they are flooded (McKean et al., 2008). Near-channel vegetation is primarily grasses and willows with occasional conifer trees and there is very little in-stream wood. Base flows after snowmelt runoff are usually reached in late July in Bear Valley Creek and average approximately 1 m³/s.

3. Methods

We mapped the channel and floodplain topography in the study reach using the airborne aquatic-terrestrial Experimental Advanced Airborne Research Lidar (EAARL; McKean et al., 2009). The EAARL is a low power, narrow-beam, full-waveform instrument with a scanning and rapidly pulsing green laser that can take measurements subaerially and through water (Bonisteel et al., 2009; Nagle & Wright, 2016). Each laser pulse measures the location and elevation of a circular ground footprint 15–20 cm in diameter and the spacing of these measurements was 1–2 m. We then interpolated this LiDAR data point cloud of 500,000 elevation measurements to define a DEM with a 1 m pixel resolution and mapped the channel and floodplain features in that DEM (SI). The uncertainty associated with the LiDAR's vertical elevation measurements has a standard deviation of 0.1 m, which has a limited impact on flow hydraulic predictions for ecohydraulics applications, for example, 0.06 standard deviation of cell suitability index and less than 0.01 m error on water surface elevations (Tonina et al., 2020).

Trends in flow for the study reach were described for the 60-year period of 1957–2016 based on correlations with historic flow records measured at eight gauges in nearby streams. These gauges had incomplete records (average number of years with data was 44) and missing years were imputed to complete the time series using an iterative Principle Components Analysis (PCA) with the MissMDA package (Missing Values with Multivariate Data Analysis; Josse & Husson, 2016) in the R statistical program (version 3.2) (R Core Team, 2019). Because there was no permanent stream gauge in the study reach, a temporary staff gauge was operated there during 2008 to define the relationship between local discharge and water surface elevation in the channel (Stage-discharge relationship, Figure S2b in Supporting Information S1).

To describe the historical summer discharge trend in the study reach, we calculated annual July–September mean monthly flow for 1957–2016 and fit a trend to the data time-series to smooth the annual variability (Stage-discharge





Figure 1. (a) Location of the study site in the northwestern U.S. (b) Floodplain topography and channel bathymetry mapped by the Experimental Advanced Airborne Research Lidar instrument. Black dots are individual LiDAR elevation measurement points. (c) Spatial pattern of flow velocity in the mainstem channel during a discharge of 1 m³/s. (d) Distribution of Chinook salmon spawning habitat quality during a discharge of 1 m³/s. See equation 3 in Supporting Information S1 for calculation of habitat quality. (e) Off-channel rearing habitat (OCH) as a function of water stages in 1957 and 2090. The two circled areas illustrate examples where water flows freely through the OCH at the higher stage (1957 conditions), but the habitat becomes fragmented and characterized by standing water at the lower stage (2090 conditions).

relationship, Figure S2a in Supporting Information S1). The estimated trend was -0.05 m^3 /s/decade, or about -3%/decade relative to the summer average monthly flow average in 1957. To derive future late summer flow values, we forced the Variable Infiltration Capacity (VIC) hydrologic model with global climate model (GCM) projections, which were biased corrected (Maurer, 2007; Wood et al., 2002) (SI: Climate-change scenarios), of air temperature and precipitation changes associated with the A1B, B1 and ensemble mean, EM, emissions scenarios during the 2050s (2030–2059) and 2090s (2070–2099; Elsner et al., 2013). Comparisons between predicted monthly VIC discharges and those measured at Bear Valley Creek showed discrepancies around 24% of the historical values (1922–1957) and less than 10% in daily discharge collected between 2006 and 2008 (VIC model validation, Figure S3 in Supporting Information S1). The three emission scenarios provided similar future results, so we selected the EM predictions for further analysis (Reeder et al., 2021) because EM predictions weight the effect of all GSMs scenarios (Climate-change scenarios comparison: Figure S5 in Supporting Information S1).

To delineate off-channel habitats important for juvenile salmon rearing, we mapped the spatial patterns of inundated floodplain topography that were connected to the mainstem channel of Bear Valley Creek during water stages corresponding to the mean trend discharges in 1957, 2040 and 2090. Mapped polygons were georeferenced by their downstream connection point to the main channel. The surface area, volume, and mean depth of each off-channel habitat polygon were calculated at discrete water discharges, using their respective water stages and the high-resolution LiDAR-derived stream bathymetry.

We used the 2D hydrodynamic Flow and Sediment Transport with Morphological Evolution of Channels (FaSTMECH; Nelson et al., 2016) model that was calibrated and validated to the study reach to quantify flow hydraulics, for example, local depth, velocity and shear stress (Carnie et al., 2015). Comparison of predicted and measured water surface elevations and velocity at low discharge $(1.6 \text{ m}^3/\text{s})$ showed good agreement with root mean square errors of 0.054 m and 0.06 m/s, respectively, which were within the error expected in the literature (Pasternack et al., 2006). We numerically computed in-channel flow characteristics at discharges between 1 and 7 m³/s at 1 m³/s increments. This covered the range in discharge values from low flows in late autumn-winter to spring snowmelt flood flows that fill the channel to the top of the streambanks. The model predicted water depth, flow velocity and shear stress near the channel bed with a 1 m² spatial resolution at each analyzed discharge (e.g., Figures S6 and S7 in Supporting Information S1). Mean velocity for the full study reach was computed from the mapped velocity distributions.

To predict in-channel spawning and rearing habitats for Chinook salmon, we coupled the results of the hydraulic model predictions with published habitat suitability curves from the Washington Department of Fish and Wildlife (Carnie et al., 2015; WDFW, 2004) that expressed the observed preferences of this species for ranges of flow depth, flow velocity, substrate grain size and percent sand. The quality of local spawning habitat, expressed as cell suitability index, CSI, was calculated at the cell scale, by using the geometric mean of the suitability index of each variable (Ecohydraulic modeling, Equation 3, Figure S8 in Supporting Information S1). We quantified the weighted useable area, WUA (SI: Ecohydraulic modeling, Equation 4), to provide a reach scale index of habitat availability (Reeder et al., 2021). The substrate grain size and sand concentrations were mapped in field surveys during 2011. Similar calculations with rearing suitability curves (WDFW, 2004) for flow depth, flow velocity and grain size distribution were made for local in-channel salmon rearing habitat quality (SI: Ecohydraulic modeling).

4. Results and Discussion

During a recent historical period of 1957–2016, average flows in the study reach declined by 19% (-3%/decade) during the summer months of July, August, and September (Figure 2a), a decrease that was consistent, although smaller, with declines observed at the majority of gauges on unregulated streams in the region during this period (Figure S4 in Supporting Information S1, Isaak et al., 2018). This flow decrease affected physical conditions within the main channel used by adults for spawning and in adjacent off-channel areas used for rearing by juve-nile salmon. The overall wetted dimensions of the main channel decreased by a small amount (-2% to -3%; Figure 2b), but the average flow velocity through the channel dropped more substantially (-17%; Figure 2a). The surface area of off-channel rearing sites also declined over the same time (-6%; Figure 2c). Importantly, the spatial distribution of these changes was not uniform. For example, the largest drops in velocity occurred within an approximately 1000 m² area composed of riffle channel habitats where adult spawning is usually concentrated.





Figure 2. Trends estimated from linear regressions for the historic period (1957–2021, blue lines with pink fill for standard error) and over the full study extent (1957–2090, black lines with gray fill for standard error) in yearly values of average summer reach conditions for Bear Valley Creek under the ensemble mean warming scenario. Reach characteristics are: (a) Discharge and flow velocity, (b) wetted area and depth, (c) off-channel habitat area, and (d) weighted useable area for adult spawning and juvenile rearing. Trends over the full study extent are statistically significant (p < 0.001), Q = -0.01 year + 21.74, V = -0.001 year + 2.37, A = -88.91 year + 284,539, D = -0.00055 year + 1.58, OCH = 71.5 year + 172,771.7, S = -130.78 year + 302,129, and R = -13.45 year + 99122.3 (year expressed as calendar year, e.g., 1957), whereas those for the historic period are not statistically significant. Shaded areas around the linear trends identify the 95% confidence level intervals.

Those changes in channel hydraulics also translated to effects on aquatic habitat conditions. For example, the WUA of good spawning habitat (CSI > 0.8) was reduced by about 10% (Figure 2d), That occurred while the habitat area for in-channel rearing of juveniles had a negligible decline (-0.1%). These differences in habitat response within the main channel reflect both a stronger dependence of good spawning sites on higher local water velocity and the larger declines in velocity at the spawning sites relative to the rearing areas. Overall, off-channel juvenile rearing habitats also declined strongly over time (-6%) as these rearing areas shrank or became disconnected at lower discharges and water stages. The off-channel habitat was also predicted to begin breaking up into smaller patches (<200 m²) in the future as average water stages decreased and large patches disappear (Figure 3).

In addition to the loss of habitat area over time, other ecologically important changes also occur at lower stream discharges. For example, off-channel rearing habitats for juvenile salmon often consist of secondary channels with flows that return to the main channel at downstream connection points (Figure 1d). As flows decline, one end of these habitats may be severed from the main channel or they may separate in the middle, causing flows to stagnate and declines in habitat quality for salmon as water temperatures increase, dissolved oxygen levels decrease (Henning et al., 2006) and macroinvertebrate communities adjust in ways that alter the potential food supply (Paillex et al., 2006, 2009). Locations of connection points along the reach depend on subtleties in geomorphic





Figure 3. Change in summer off-channel rearing habitat, off-channel habitats (OCH), size frequency distributions between historical, 1957, and future, 2040 and 2090, periods when the mean water stage height is predicted to decline by 10 cm. Whiskers represents OCH variability due to 24% discharge uncertainty quantified for future predictions. The symmetrical uncertainty of the discharge results in unsymmetric whiskers around the expected OCH value because OCH extension is due to the interaction between discharge stage and off-channel topography.

floodplain characteristics but are often spatially clustered (McKean et al., 2008). Once critical low-flow thresholds are exceeded, some sub-reaches experience large local habitat losses and fragmentation of networks despite the total wetted surface area remaining relatively constant. Our estimates suggest about 20% of the off-channel rearing habitat with through-flowing water was lost between 1957 and 2016 (Figure 2d).

Observed decreasing historical trends in wetted channel dimensions and salmon habitats are predicted to continue and intensify later this century under the EM scenario, with mean summer flows estimated to decline by about 45% in 2040 and 72% in 2090 relative to the 1957 baseline. If these changes transpire, the amount of main-channel habitat for juveniles is again predicted to show relatively small changes compared to the decreases in spawning (-38%) and off-channel rearing habitats (-29%; Figures 2c and 2d). Besides shrinking, the reduction in mean summer discharge from nearly 2 m³/s historically (1957) to 1 m³/s around 2040 is predicted to decrease the size of habitat patches (average habitat size reduces by 46% from 84 to 46 m², respectively) (Table S1 in Supporting Information S1). The spawning habitat also became more fragmented as the number of habitat patches with less than the minimum patch size of 6 m² (corresponding to the average surface area of a salmon nest) increased by 95% compared to larger habitats of good and excellent quality (Table S1 in Supporting Information S1). About 30%–35% percent of the remaining off-channel habitat with flowing water would also be fragmented and reduced to areas of standing water. Uncertainty on OCH increases with OCH extension (low uncertainty for small OCH size) but it does not mask a systematic reduction and or disappearance of large contiguous OCH patches (Figure 3).

Summer flow reductions are common to many snow-dependent regions (Stewart, 2009) suggesting detrimental effects not only to the Chinook salmon considered here, but also to other species of salmon, trout and char that require cold water from snowmelt runoff and are regulated at multiple life stages by density-dependent growth, survival, and reproduction (Quinn, 2005). As rearing juveniles, the fish organize into spatial hierarchies where dominant individuals occupy bioenergetically profitable positions to maximize growth opportunities. Similar territorial mechanisms occur at the adult life stage when dominant females establish breeding territories in the best habitats to build nests and males compete aggressively for access to the females. Even at contemporary abundance levels that are reduced from historical numbers, these density-dependent processes often manifest strongly

at local scales (Achord et al., 2003). Habitat reductions are especially problematic during the summer low-flow season when bioenergetic expenditures are greatest to take advantage of limited seasonal growth opportunities or to cope with physiological stresses in warm environments. Biological responses may take one of several forms, including direct mortality caused by increasing female superimposition of nests when spawning is confined to smaller useable habitat areas (von Biela et al., 2022), reduced juvenile growth rates, or altered juvenile emigration rates and timing (Holtby, 1988). Reductions in juvenile size-at-age may contribute to higher mortality rates (Quinn, 2005) and forced emigration is also usually disadvantageous. Whereas warmer water temperatures may sometimes increase growth because of high metabolic rates if sufficient food is also available; higher fish densities caused by a decrease in rearing habitat could negate the effect (Reeder et al., 2021).

The VIC model forced by several GCMs and emissions scenarios predicted similar future hydrographs with mean summer discharges declining (-6%/decade) at faster rates than historic (-3%/decade) values within Bear Valley Creek but comparable to historic summer flow reductions in nearby unregulated streams that ranged between -4 and -6%/decade (VIC model validation, Figure S4 in Supporting Information S1). Residuals between VIC discharge predictions and values measured during the historic period are also constrained within 24% with a marginal error of 0.27 m³/s, which is smaller than the change in flow prediction over several decades. An important assumption in our study is that we considered the streambed and floodplain morphologies fixed over time (Reid et al., 2020). Whereas previous research predicts limited streambed mobility in this stream over a range of flood sizes (McKean and Tonina, 2013), changes in hillslope processes with increased potential for wildfires and fine sediment delivered to the stream could increase the deposition of fine sediment (Sankey et al., 2017). This would negatively impact spawning habitat by clogging flow through nest sites, which may reduce embryo survival by decreasing oxygen delivery and waste removal (Zimmermann & Lapointe, 2005; Tonina & Buffington, 2009). Increased fine loading could also cause development of natural levees (Maturana et al., 2013) that inhibit connection with lateral channels and reduce the available OCH area. Consequently, our calculations of the impacts of climate change on future salmon habitat quality may be conservative.

Climate-induced reductions in the regional capacity of freshwater habitats to support salmon populations or traditional life histories pose unanticipated challenges for conservation strategies that assume reversals of past habitat degradation will be sufficient to ensure long-term population persistence (Kareiva et al., 2000). Toward that end, billions of US\$ have been spent on habitat restoration and population conservation efforts in the northwestern U.S. while the biological benefits remain subject to debate (Rieman et al., 2015). Nonetheless, costly investments will continue this century, so efforts to improve investment strategies should be ongoing and can benefit from more precise, spatially explicit information about the biological consequences of habitat improvements and climate trends. To achieve better river analyses and management, we must also survey or model many biophysical aspects of stream systems. At a minimum, these include channel and floodplain topography, water depth, temperature and velocity, the grain size of sediment on the stream bed and the location of logs and other large woody debris that force small scale hydraulics as well as provide local habitat. Furthermore, stream biota conduct their daily activities in microhabitats with a scale of a few meters, but highly mobile species like anadromous fish also use thousands of kilometers of rivers during their migratory life stages. Thus, we need to accurately describe relevant biological phenomena such as distributions, growth, and productivity at high-resolution over a comparable range of spatial scales to facilitate strategic assessments of habitat use and means of restoration (Torgersen et al., 2021). This is difficult, but achievable by exploiting recent advances using new technologies such as bathymetric lidars and drone-based optical surveys and improved numerical models of channel hydrodynamics.

5. Conclusions

Climate change is not only impacting stream water temperature regimes in high elevation spawning and rearing meadow systems, like Bear Valley Creek, but is also affecting hydrological regimes. Expected future summer flows will be lower than historic (\sim -72%), which will cause continued reductions in and fragmentation of spawning (\sim -38%) and rearing (\sim -2.5%) and off-channel (\sim -29%) habitats by the end of the century. These predicted negative impacts on habitat extent and distribution will be potentially exacerbated by an increase of late summer water temperature and potentially by deposition of fine sediment, which may also further disconnect the main stem from off-channel habitats. This detailed analysis also demonstrates that new technologies, able to define and analyze continuous physical conditions of the stream and floodplain at the micro-habitat scale over long stream domains, can improve riverine ecosystem analyses and management.

Conflict of Interest

The authors declare no conflicts of interest relevant to this study.

Data Availability Statement

All data are available in the supporting material or stored in HydroShare repository (Tonina et al., 2020, Hydro-Share, http://www.hydroshare.org/resource/952304525d0a450b93842ce5068b2d11).

References

- Achord, S., Levin, P. S., & Zabel, R. W. (2003). Density-dependent mortality in Pacific salmon: The ghost of impacts past? *Ecology Letters*, 6(4), 335–342. https://doi.org/10.1046/j.1461-0248.2003.00438.x
- Bonisteel, J. M., Nayegandhi, A., Wright, C. W., Brock, J. C., & Nagle, D. B. (2009). Experimental advanced airborne research lidar (EAARL) data processing manual. U.S. Geological Survey. https://doi.org/10.3133/ofr20091078
- Carbonneau, P. E., Fonstad, M. A., Marcus, W. A., & Dugdale, S. J. (2012). Making riverscapes real. Geomorphology, 137(1), 74–86. https://doi. org/10.1016/j.geomorph.2010.09.030
- Carnie, R., Tonina, D., McKean, J. A., & Isaak, D. J. (2015). Habitat connectivity as a metric for aquatic microhabitat quality: Application to Chinook Salmon spawning habitat. *Ecohydrology*, 9(6), 982–994. https://doi.org/10.1002/eco.1696
- Dudgeon, D., Arthington, A. H., Gessner, M. O., Kawabata, Z., Knowler, D. J., Lévêque, C., et al. (2006). Freshwater biodiversity: Importance, threats, status and conservation challenges. *Biological Review*, 81(2), 163–182. https://doi.org/10.1017/s1464793105006950
- Elsner, M. M., Mauger, G. S., Lee, S.-Y., Tohver, I., & Norheim, R. A. (2013). An overview of the Columbia Basin climate change scenarios project: Approach, methods, and summary of key results. *Atmosphere-Ocean*, 51(4), 392–415. https://doi.org/10.1080/07055900.2013. 819555
- Henning, J. A., Gresswell, R. E., & Fleming, I. A. (2006). Juvenile salmonid use of freshwater emergent wetlands in the floodplain and its implications for conservation management. North American Journal of Fisheries Management, 26(2), 367–376. https://doi.org/10.1577/m05-057.1

Holtby, L. B. (1988). Effects of logging on stream temperatures in Carnation Creek British Columbia, and associated impacts on the Coho Salmon (Oncorhynchus Kisutch). Canadian Journal of Fisheries and Aquatic Sciences, 45(3), 502–515. https://doi.org/10.1139/f88-060

- Isaak, D. J., Luce, C. H., Horan, D. L., Chandler, G., Wollrab, S., & Nagel, D. (2018). Global warming of salmon and trout rivers in the northwestern U.S: Road to ruin or path through purgatory? *Transactions of the American Fisheries Society*, 147(3), 566–587. https://doi.org/10.1002/ tafs.10059
- Josse, J., & Husson, F. (2016). MissMDA: A package for handling missing values in multivariate data analysis. *Journal of Statistical Software*, 70, 1–31. https://doi.org/10.18637/jss.v070.i01
- Kareiva, P., Marvier, M., & McClure, M. (2000). Recovery and management options for Spring/summer Chinook Salmon in the Columbia River Basin (Vol. 290(5493), pp. 977–979).
- Kormos, P., Luce, C. H., Wegner, S. J., & Berghuijs, W. R. (2016). Trends and sensitivities of low streamflow extremes to discharge timing and magnitude in Pacific Northwest mountain streams. *Water Resources Research*, 52(7), 4990–5007. https://doi.org/10.1002/2015wr018125
- Lewis, B., Grant, W. S., Brenner, R. E., & Hamazaki, T. (2015). Changes in size and age of Chinook salmon Oncorhynchus Tshawytscha returning to Alaska. PLOS ONE, 10(6), e0130184. https://doi.org/10.1371/journal.pone.0130184
- Maturana, O., Tonina, D., McKean, J. A., Buffington, J. M., Luce, C. H., & Caamaño, D. (2013). Modeling the effects of pulsed versus chronic sand inputs on salmonid spawning habitat in a low-gradient gravel-bed river. *Earth Surface Processes and Landforms*, 39(7), 877–889. https:// doi.org/10.1002/esp.3491
- Maurer, E. P. (2007). Uncertainty in hydrologic impacts of climate change in the Sierra Nevada, California, under two emissions scenarios. *Climatic Change*, 82(3–4), 309–325. https://doi.org/10.1007/s10584-006-9180-9
- McKean, J. A., & Tonina, D. (2013). Bed stability in unconfined gravel bed mountain streams: With implications for salmon spawning viability in future climates. *Journal of Geophysical Research: Earth Surface*, 118(3), 1227–1240. https://doi.org/10.1002/jgrf.20092
- McKean, J. A., Isaak, D. J., & Wright, C. W. (2008). Geomorphic controls on salmon nesting patterns described by a new, narrow-beam terrestrial-aquatic lidar. Frontiers in Ecology and the Environment, 6(3), 125–130. https://doi.org/10.1890/070109
- McKean, J. A., Nagel, D., Tonina, D., Bailey, P., Wright, C. W., Bohn, C., & Nayegandhi, A. (2009). Remote sensing of channels and riparian zones with a narrow-beam aquatic-terrestrial LiDAR. *Remote Sensing*, 1(4), 1065–1096. https://doi.org/10.3390/rs1041065
- Nagle, D. B., & Wright, C. W. (2016). Algorithms used in the airborne lidar processing system (ALPS): U.S. Geological survey open-file report. (Vol. 45). US Geological Survey.
- Naman, S. M., Rosenfeld, J. S., & Richardson, J. S. (2016). Causes and consequences of invertebrate drift in running waters: From individuals to populations and trophic fluxes. *Canadian Journal of Fisheries and Aquatic Sciences*, 73(8), 1292–1305. https://doi.org/10.1139/ cjfas-2015-0363
- Nehlsen, W., Williams, J. E., & Lichatowich, J. A. (1991). Pacific salmon at the crossroads: Stocks at risk from California, Oregon, Idaho, and Washington. *Fisheries*, *16*(2), 4–21. https://doi.org/10.1577/1548-8446(1991)016<0004:psatcs>2.0.co;2
- Nelson, J. M., Shimizu, Y., Abe, T., Asahi, K., Gamou, M., Inoue, T., et al. (2016). The international river interface cooperative: Public domain flow and morphodynamics software for education and applications. Advances in Water Resources, 93(Part A), 62–74. https://doi.org/10.1016/j. advwatres.2015.09.017
- Ohlberger, J., Buehrens, T. W., Brenkman, S. J., Crain, P., Quinn, T. P., & Hilborn, R. (2018). Effects of past and projected river discharge variability on freshwater production in an anadromous fish. *Freshwater Biology*, 63(4), 331–340. https://doi.org/10.1111/fwb.13070
- Paillex, A., Castella, E., & Carron, G. (2006). Aquatic macroinvertebrate response along a gradient of lateral connectivity in river floodplain channels. Journal of the North American Benthological Society, 26(4), 779–796. https://doi.org/10.1899/06-12.1
- Paillex, A., Doledec, S., Castella, E., & Merigoux, S. (2009). Large river floodplain restoration: Predicting species richness and trait responses to the restoration of hydrological connectivity. *Journal of Applied Ecology*, 46(1), 250–258. https://doi.org/10.1111/j.1365-2664.2008.01593.x
- Parmesan, C., & Yohe, G. (2003). A globally coherent fingerprint of climate change impacts across natural systems. *Nature*, 421(6918), 37–42. https://doi.org/10.1038/nature01286

Acknowledgments

We thank Frank Gariglio, Jeff Reader, William Jeff Reeder, Heidi Schott and Amar Nayegandhi for their efforts in data collection. We thank two anonymous reviewers and the Editor, Dr. Valeriy Ivanov, for their critical, valuable, and constructive suggestions and comments, which improved the contribution. DT, DI and JM are grateful to the U.S. Forest Service, Rocky Mountain Research Station for project support and QC is funded by the National Natural Science Foundation of China (No. 52121005, 92047303). Any opinions, conclusions, or recommendations expressed in this material are those of the authors and do not necessarily reflect the views or the policy of the supporting agencies, the U.S. Environmental Protection Agency and the U.S. Forest Service.

- Pasternack, G. B., Gilbert, A. T., Wheaton, J. M., & Buckland, E. M. (2006). Error propagation for velocity and shear stress prediction using 2D models for environmental management. *Journal of Hydrology*, 328(1–2), 227–241. https://doi.org/10.1016/j.jhydrol.2005.12.003
- Potter, K. A., Arthur Woods, H., & Pincebourde, S. (2013). Microclimatic challenges in global change biology. Global Change Biology, 19(10), 2932–2939. https://doi.org/10.1111/gcb.12257
- Quinn, T. P. (2005). The behavior and ecology of Pacific Salmon and trout, University of Washington Press.
- R Core Team. (2019). R: A language and environment for statistical computing. R Foundation for Statistical Computing. Retrieved from https://www.R-project.org/
- Reeder, W. J., Gariglio, F., Carnie, R., Tang, C., Isaak, D., Chen, Q., et al. (2021). Some (fish might) like it hot: Habitat quality and fish growth from past to future climates. *The Science of the Total Environment*, 787, 147532. https://doi.org/10.1016/J.SCITOTENV.2021.147532
- Reid, D. A., Hassan, M. A., Bird, S., Pike, R., & Tschaplinski, P. (2020). Does variable channel morphology lead to dynamic Salmon habitat? *Earth Surface Processes and Landforms*, 45(2), 295–311. https://doi.org/10.1002/esp.4726
- Rieman, B. E., Smith, C. L., Naiman, R. J., Ruggeronee, G. T., Wood, C. C., Huntly, N., et al. (2015). A comprehensive approach for habitat restoration in the Columbia Basin. *Fisheries*, 40(3), 124–135. https://doi.org/10.1080/03632415.2015.1007205
- Sankey, J. B., Kreitler, J., Hawbaker, T. J., McVay, J. L., Miller, M. E., Mueller, E. R., et al. (2017). Climate, wildfire, and erosion ensemble foretells more sediment in western USA watersheds. *Geophysical Research Letters*, 44(17), 8884–8892. https://doi.org/10.1002/2017GL073979
- Schmidt, D. L., & Mackin, J. H. (1970). Quaternary geology of long and bear valleys, West-Central Idaho (Vols. 1211-A). Geological Survey Bulletin.
- Stewart, I. T. (2009). Changes in snowpack and snowmelt runoff for key mountain regions. *Hydrological Processes*, 23(1), 78–94. https://doi.org/10.1002/hyp.7128
- Tonina, D., & Buffington, J. M. (2009). A three-dimensional model for analyzing the effects of salmon redds on hyporheic exchange and egg pocket habitat. *Canadian Journal of Fisheries and Aquatic Sciences*, 66(12), 2157–2173. https://doi.org/10.1139/f09-146
- Tonina, D., McKean, J. A., Benjankar, R. M., Yager, E., Carmichael, R. A., Chen, Q., et al. (2020). Evaluating the performance of topobathymetric LiDAR to support multi-dimensional flow modelling in a gravel-bed mountain stream. *Earth Surface Processes and Landforms*, 45(12), 2850–2868. https://doi.org/10.1002/esp.4934
- Torgersen, C. E., Le Pichon, C., Fullerton, A. H., Dugdale, S. J., Duda, J. J., Giovannini, F., et al. (2021). Riverscape approaches in practice: Perspectives and applications. *Biological Reviews*, 97(2), 481–504. https://doi.org/10.1111/brv.12810
- Urban, M. C. (2015). Accelerating extinction risk from climate change. *Science*, 348(6234), 571–573. https://doi.org/10.1126/science.aaa4984 von Biela, V. R., Sergeant, C. J., Carey, M. P., Liller, Z., Russell, C., Quinn-Davidson, S., et al. (2022). Premature mortality Observations among
- Alaska's Pacific Salmon during record heat and drought in 2019. Fisheries, 47, 157–168. https://doi.org/10.1002/fsh.10705
 WDFW. (2004). Instream flow study guidelines: Technical habitat suitability issues including fish preference curves, Washington Department of Fish and Wildlife.
- Wenger, S. J., Isaak, D. J., Luce, C. H., Neville, H. M., Fausch, K. D., Dunham, J. B., et al. (2011). Flow regime, temperature, and biotic interactions drive differential declines of trout species under climate change. *Proceedings of the National Academy of Sciences of the United States of America*, 108(34), 14175–14180. https://doi.org/10.1073/pnas.1103097108
- Wood, A. W., Maurer, E. P., Kumar, A., & Lettenmaier, D. P. (2002). Long-range experimental hydrologic forecasting for the eastern United States. Journal of Geophysical Research, 107(20), 4429. https://doi.org/10.1029/2001JD000659
- Young, M. K., Isaak, D. J., Spaulding, S., Thomas, C. A., Barndt, S. A., Groce, M. C., et al. (2018). Climate vulnerability of native cold-water salmonids in the northern rockies region [chapter 5]. In J. E., Halofsky, D. L., Peterson, S. K., Dante-Wood, L., Hoang, J. J., Ho, & L. A., Joyce (Eds.), *Climate change vulnerability and adaptation in the northern Rocky Mountains [Part 1]*. Gen. Tech. Rep. RMRS-GTR-374 (pp. 87–127). U.S. Department of Agriculture, Forest Service, Rocky Mountain Research Station.
- Zimmermann, A. E., & Lapointe, M. F. (2005). Intergranular flow velocity through salmonid redds: sensitivity to fines infiltration from low intensity sediment transport events. *River Research and Applications*, 21(8), 865–881.

References From the Supporting Information

- Cobby, D. M., Mason, D. C., & Davenport, I. J., (2001). Image processing of airborne scanning laser altimetry data for improved river flood modelling. ISPRS Journal of Photogrammetry and Remote Sensing, 56(2), 121–138. https://doi.org/10.1016/S0924-2716(01)00039-9
- Hamman, J. J., Nijssen, B., Bohn, T. J., Gergel, D. R., & Mao, Y. (2018). The variable infiltration capacity model version 5 (VIC-5): Infrastructure improvements for new applications and reproducibility. *Geoscientific Model Development*, 11(8), 3481–3496. https://doi.org/10.5194/ gmd-11-3481-2018
- IPCC IP on CC. (2007). Climate change and water. Intergovernmental Panel on Climate Change Technical Paper VI, IPCC Secretariat.
- Legleiter, C. J., & Kyriakidis, P. C. (2008). Spatial prediction of river channel topography by kriging. Earth Surface Processes and Landforms, 33, 841–867.
- Liang, X., Lettenmaier, D. P., Wood, E. F., & Burges, S. J. (1994). A simple hydrologically based model of land-surface water and energy fluxes for general-circulation models. *Journal of Geophysical Research-Atmospheres*, 99, 14415–14428.
- Mantua, N., Tohver, I., & Hamlet, A. (2010). Climate change impacts on streamflow extremes and summertime stream temperature and their possible consequences for freshwater salmon habitat in Washington State. *Climatic Change*, 102(1–2), 187–223. https://doi.org/10.1007/ s10584-010-9845-2
- Maraun, D., Shepherd, T. G., Widmann, M., Zappa, G., Walton, D., Gutiérrez, J. M., et al. (2017). Towards process-informed bias correction of climate change simulations. *Nature Climate Change*, 7(11), 764–773. https://doi.org/10.1038/nclimate3418
- McKean, J. A., Tonina, D., Bohn, C., & Wright, C. W. (2014). Effects of bathymetric LiDAR errors on flow properties predicted with a multi-dimensional hydraulic model. *Journal of Geophysical Research: Earth Surface 119*, (3), 644–664. https://doi.org/10.1002/2013JF002897
- McDonald, R. R., Nelson, J. M., & Bennett, J. P., (2005). Multi-dimensional surface-water modeling system user's guide, U.S. Geological Survey. Mote, P. W., & Salathé, E. P. (2010). Future climate in the Pacific Northwest. Climatic Change, 102(1–2), 29–50. https://doi.org/10.1007/ s10584-010-9848-z
- Qiu, L., Kim, J.-B., Kim, S.-H., Choi, Y.-W., Im, E.-S., & Bae, D.-H., (2022). Reduction of the uncertainties in the hydrological projections in Korean river basins using dynamically downscaled climate projections. *Climate Dynamics*, https://doi.org/10.1007/s00382-022-06201-8
- Todini, E. (1996). The ARNO rainfall-runoff model. Journal of Hydrology, 175(1–4), 339–382. https://doi.org/10.1016/S0022-1694(96)80016-3
 Tonina, D., & Jorde, K. (2013). Hydraulic Modelling Approaches for Ecohydraulic Studies: 3D, 2D, 1D and Non-Numerical Models. In Ecohydraulics: An Integrated Approach 31–74. https://doi.org/10.1002/9781118526576.ch3



Wright, A. N., Schwartz, M. W., Hijmans, R. J., & Shaffer, H. B. (2015). Advances in climate models from CMIP3 to CMIP5 do not change wingin, in 14, behaving in 14, in Fightans, it 5, to online, in *D*. (2015). Invalues in clinic indexes from clinic of the child of the

Proceedings of the National Academy of Sciences, 104(38), 14889-14894. https://doi.org/10.1073/pnas.0701020104